Author's response to reviews

Title: SNP Microarray analyses reveal copy number alterations and progressive genome reorganization during tumor development in mice breast cancer

Authors:

Christoph Standfuß (christoph.standfuss@th-wildau.de)
Heike Pospisil (pospisil@th-wildau.de)
Andreas Klein (andreas.klein@charite.de)

Version: 2 Date: 28 February 2012

Author's response to reviews:

Dear Dr. Chap,

we noticed that two figures were not displayed correctly. That is why, we added two new figures (figure 1 and figure 5).

Sincerely,
Heike Pospisil